Author's response to reviews

Title: A trial for the use of qigong in the treatment of pre and mild essential hypertension: a study protocol for a randomized controlled trial

Authors:

Ji-Eun Park (jepark@kiom.re.kr)
Yan Liu (liuyan@kiom.re.kr)
Taeseob Park (ptsissh@hanmail.net)
Sanghoon Hong (shhong@deu.ac.kr)
Jung-Eun Kim (eujuki@kiom.re.kr)
Tae-Hun Kim (neijing@kiom.re.kr)
Ae-Ran Kim (arkim@kiom.re.kr)
So-Young Jung (syjung@kiom.re.kr)
Hyoju Park (mable@kiom.re.kr)
Sun-Mi Choi (smchoi@kiom.re.kr)

Version: 3 Date: 17 November 2011

Author's response to reviews: see over
Answer for Editorial reviewers

Thanks for your helps and suggestions. The point to point responses to the comments of the editorial reviewer are listed below. We checked the changed sentence in red color.

Editorial reviewers' comments:

Tables: We notice you have included a list as table 1. Please reformat so it is incorporated within the main text. Please remember to update any citations to it in the text.
-> Main text incorporated the table 1. We modified it to be identical. (page 5, 16)

Figure legends: please provide figure titles and legends in a section entitled 'Figure legends' directly after the reference list. Figure titles should be concise and be the first sentence of the figure legend. Figure Legends should be no more than 300 and should describe what is illustrated in the manuscript.
-> We modified 'Figure legends' to be directly after the reference list. Figure legends are less than 300 and it is illustrated in manuscript. (page 4, 15)

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
-> We checked the manuscript. Thank you.

Thank you for all valuable comments.